Author's response to reviews

Title: Evaluation of potential drug-herb interactions among a group of Palestinian patients with chronic diseases

Authors:

Rowa Al Ramahi (rawa_ramahi@najah.edu)
Nidal Jaradat (nidaljaradat@yahoo.com)
Rubia Shalalfeh (lara_alkhatib@yahoo.com)
Sojoud Nasir (mutamed@yahoo.com)
Yazan Manasra (rowa_ramahi@gmail.com)
Ihab Shalalfeh (sajed_alkhatib@yahoo.com)
Yasmen Esam (rowa_ramahi@yahoo.com)

Version: 2  Date: 23 September 2014

Author's response to reviews: see over
Dear Editor,

Changes made:

Line and page numbers have been added.

In the Methods section of the manuscript, the name of the body which gave approval has been added.

In the Methods section, it has been stated that an oral informed consent for participation in the study was obtained from participants.

Authors' contributions have been added.

Acknowledgements have been added.

I could add the e mails of the co-authors.

Title Page has been moved to the beginning of manuscript document.

Thanks for your time

With best regards

Rowa